Author’s response to reviews

Title: Clinical course, mutations and its functional characteristics of infantile-onset Pompe disease in Thailand

Authors:

Lukana Ngiwsara (lukana@cri.or.th)
Duangrudee Wattanasirichaigoon (duangrudee.wat@mahidol.ac.th)
Thipwimol Tim-Aroon (thipwimol.tim@mahidol.ac.th)
Kitiwan Rojneuangnit (kitiw@n@gmail.com)
Saisuda Noojaroen (fernsaisuda@gmail.com)
Arthaporn Khongkraron (manualsibc@gmail.com)
Phannee Sawangareetrakul (Phannee@cri.or.th)
James R. Ketudat-Cairns (cairns@sut.ac.th)
Ratana Charoenwattanasatien (na_ratana@hotmail.com)
Voraratt Champattanachai (Voraratt@cri.or.th)
Chulaluck Kuptanon (chulaluck.kuptanon@gmail.com)
Suthipong Pangkanon (suthipongpangkanon@gmail.com)
Jisnuson Svasti (jisnuson.sva@mahidol.ac.th)

Version: 6 Date: 20 Aug 2019

Author’s response to reviews:

Dear The editors

We are very much grateful of receiving another opportunity to revise the manuscript.

We have made changes following your instructions 1, 2, 3 as follow.
1. Title page

Response: We removed the statement "Phone: +66 22012782; Fax: +66 22012783 Mobile: +66 814003957" from the title page in your manuscript.

2. Ethics approval and consent to participate

Response: The statement is changed to “Written informed consent was obtained from all participants and the parents of the minor participants, after ethics approval was obtained from the Ramathibodi Hospital Institutional Review Board (protocol ID 06-55-46).”

3. Availability of data and materials

Response: The statement is changed to “The datasets used and/or analyzed during the current study are available from the corresponding author upon request and on the ClinVar website (https://www.ncbi.nlm.nih.gov/clinvar/) using the ClinVar accession numbers provided in Table 3.”

4. Authors’ contributions

Please amend the initials of Thipwimol Tim-Aroon from “TT” to “TTA” in the Authors’ contributions section in your manuscript.

Response: Dr. Thipwimol Tim-Aroon confirmed that she only used “TT” and preferred to do so. Therefore, in the Authors’ contributions section, we keep it as “TT”.

Best regards

Duangrurdee